Author's response to reviews

Title: Absence of autoantibodies in girls and women with Turner syndrome

Authors:

Annika E Stenberg (annika.stenberg@ki.se)
Lisskulla Sylvén (lisskulla.sylven@karolinska.se)
Hakan Hedstrand (hakan.hedstrand@medsci.uu.se)
Olle Kampe (olle.kampe@medsci.uu.se)
Malou Hultcrantz (malou.hultcrantz@karolinska.se)

Version: 2 Date: 17 October 2007

Dear Editor

Thank you for reviewing the manuscript "Absence of autoantibodies in girls and women with Turner syndrome" by Annika E. Stenberg, Lisskulla Sylvén, Håkan Hedstrand, Olle Kämpe and Malou Hultcrantz for publication in Journal of Negative results in Biomedicine.

The comments from the reviewers have been very helpful and all has been considered carefully and corrected accordingly.

Comments to the reviewers:

Reviewer nr 1:

- The sentence has been changed accordingly.
- The table has been change to make a clearer view of the antibodies prevalent in Addison, APS I and II patients.

Reviewer nr 2:

- The title has been changed.
- All minor revisions have been changed, except for ref # 12, according to PubMed this is the right page nr.
- A short part in the discussion has been added dealing with the effect of hypothyroidism and hearing.
Yours sincerely

Annika Elmqvist Stenberg, MD, PhD

ENT Dept

Karolinska Hospital

S-171 76 Stockholm, Sweden

Tel.: +46-8-5177 50 29

Fax.: +46-8-34 85 46

E-mail: Annika.Stenberg@cns.ki.se